Author’s response to reviews

Title: The Use of a Multiple Imputation Method to Investigate the Trends in Histologic Types of Lung Cancer in Songkhla Province, Thailand, 1989-2013

Authors:

Hutcha Sriplung (hutchas@gmail.com)
Seesai Yeesoonsang (seesaiy@gmail.com)
Edward McNeil (edward.m@psu.ac.th)
Surichai Bilheem (surichai.bil@gmail.com)

Version: 2 Date: 06 Jun 2016

Author’s response to reviews:

Acknowledgements

We would like to thank the Songkhla cancer registry for providing information pertaining to lung cancer. Dr. Sriplung acknowledges the support from the National Science and Technology Development Agency, and the National Research University Grant, Prince of Songkla University as the funding sources.

Funding

National Science and Technology Development Agency (P-10-10307)
National Research University Grant, Prince of Songkla University (MED580635S)

Authors’ contribution

HS and SY conceived and designed this study. SY and SB analyzed the data. HS, SY, and EM wrote the manuscript. HS, SY and EM contributed with critical review of data analyses, interpretation of findings and critical edit of the manuscript. All authors have read and approved the final manuscript.

Competing interests
All the authors declare that they have no competing interests.

Ethics approval and consent to participate

This research was approved by the research ethics committee of the Faculty of Medicine, Prince of Songkla University, approve #58-013-18-1.